**Author's response to reviews**

**Title:** Men in Macau SAR have higher prevalence in Metabolic Syndrome and among related metabolic components: cross-sectional Macau Health Survey.

**Authors:**

Ken Gu (kengu@ipm.edu.mo)
Dulce Maia Trindade (dulmaia@gmail.com)
Qiuang Xiang Lao (xqlao@cuhk.edu.hk)
Martin Wong (drwong_martin@yahoo.com.hk)
Tong Ka Io (kitong@ssm.gov.mo)
Tanja Sobko (tanja.sobko@gmail.com)

**Version:** 2  **Date:** 27 May 2014

**Author's response to reviews:**

Revisions required prior to editorial assessment - completed

1. We have include the email address of all the authors in the title page
2. Section 'Competing interest' added.
3. Section Author's contributions have been added.

Thank you,

Kind regards,

Tanja